Author's response to reviews

Title: Association of Trypanosoma cruzi infection with risk factors and electrocardiographic abnormalities in Northeast Mexico

Authors:

Zinnia J Molina-Garza (molinazinnia@hotmail.com)
Jose L Rosales-Encina (rosales@cinvestav.mx)
Roberto Mercado-Hernandez (romercad@gmail.com)
Daniel P Molina-Garza (dpmolinagza@gmail.com)
Ricardo Gomez-Flores (ricardo.gomezf@uanl.mx)
Lucio Galaviz-Silva (lucio.galavizsl@uanl.edu.mx)

Version: 2
Date: 30 November 2013

Author's response to reviews:

Philippa Harris
Executive Editor
BMC Infectious Diseases

Ref. #: MS: 5132517210158063

Dear Philippa Harris:

I would like to thank the reviewers for their careful and detailed critique of our manuscript entitled “Association of Trypanosoma cruzi infection with risk factors and electrocardiographic abnormalities in Northeast Mexico”. I have either replied to each specific comment addressed by the reviewers, or have incorporated their suggestions in the revised manuscript. I hope that this revision will answer the thoughtful and significant concerns of the reviewers.

Sincerely,